Author’s response to reviews

Title: Comparative efficacy of three pediculicides to treat head lice infestation in primary school girls: A randomised controlled assessor blind trial in rural Iran

Authors:

Hadi Kalari (dpt2371990396@sums.ac.ir)

Aboozar Soltani (soltani_ab@sums.ac.ir)

Kourosh Azizi (azizik@sums.ac.ir)

Hossain Faramarzi (hossainfaramarzi@yahoo.com)

Mohammad Djaefar Moemenbellah-Fard (momenbf@sums.ac.ir)

Version: 2 Date: 07 Aug 2019

Author’s response to reviews:

Thanks for reviews. I have added all the stated recommendations in the new text as attached.